BYDLIŃSKI, Arkadiusz, BROŻYNA, Aleksandra, MAŁEK, Natalia, EMERLA, Sara, KWIATKOWSKA, Anita, KARŁOWICZ, Konrad, HERMANOWSKA, Maria, LUBOMIRSKA, Julia, FIGUROWSKA, Patrycja, CIULKIEWICZ, Łukasz and PLUTA, Patryk. A review of treatment of relapsing-remitting type of multiple sclerosis (RRMS). Journal of Education, Health and Sport. 2024;71:51250. eISSN 2391-8306. https://dx.doi.org/10.12775/JEHS.2024.71.51250

https://apcz.umk.pl/JEHS/article/view/51250

The journal has had 40 points in Minister of Science and Higher Education of Poland parametric evaluation. Annex to the announcement of the Minister of Education and Science of 05.01.2024 No. 32318. Has a Journal's Unique Identifier: 201159. Scientific disciplines assigned: Physical culture sciences (Field of medical and health sciences): Health Sciences (Field of medical and health sciences): Punkty Ministeriane 40 punktów. Załącznik do komunikatu Ministra Nauki i Szkolnictwa Wyższego z dnia 05.01.2024 Lp. 32318. Posiada Unikatowy Identyfikator Czasopisma: 201159. Przypisane dyscypliny naukowe: Nauki o kulture fryzyczej (Dizdzian nauk medycznych i nauk o zdrowiu); Nauki o zdrowiu, Dizdzidzian nauk medycznych i nauko vezicationa unak orgen science Unikatu Sciences University in Torun, Poland Open Access. This article is distributed under the terms of the Creative Commons Attribution Noncommercial License which permits any noncommercial license Share alike. (http://creativecommons.org/licenses/by-nc-sa/4.0) which permits unrestricted, non commercial use, distribution in any medium, provided the original author (s) and source are credited. This is an open access article licensed under the terms of the Creative Commons.org/licenses/by-nc-sa/4.0) which permits unrestricted, non commercial License duries declare that there is no conflict of interests regarding the publication of this paper. Received: 11.04.2024. Revised: 10.05.2024. Accepted: 22.05.2024. Published: 23.05.2024.

A review of treatment of relapsing-remitting type of multiple sclerosis (RRMS)

Arkadiusz Bydliński*, student Medical University of Warsaw, Żwirki i Wigury 61, 02-091 Warsaw, Poland https://orcid.org/0009-0001-4230-661X, bydlinskiarkadiusz@gmail.com Aleksandra Brożyna*, student Medical University of Warsaw, Żwirki i Wigury 61, 02-091 Warsaw, Poland https://orcid.org/0009-0000-9403-6212, ola.brozyna@icloud.com Natalia Małek, MD Central Clinical Hospital in Warsaw, Banacha 1a, 02-097 Warsaw, Poland https://orcid.org/0009-0005-9602-2929, n.malek2609@gmail.com Sara Emerla, student Medical University of Warsaw, Żwirki i Wigury 61, 02-091 Warsaw, Poland https://orcid.org/0009-0007-2229-9145, emerlasara@gmail.com Anita Kwiatkowska, MD Military Institute of Medicine - National Research Institute, Szaserów 128, 04-141 Warsaw, Poland https://orcid.org/0009-0009-7250-6194, aw.kwiatkowska@gmail.com Konrad Karłowicz, MD Central Clinical Hospital in Warsaw, Banacha 1a, 02-097 Warsaw, Poland https://orcid.org/0009-0008-4610-6456, konrad.karlowicz@uckwum.pl Maria Hermanowska, student Jan Kochanowski University, Collegium Medicum, al. IX Wieków Kielc 19A, 25-317 Kielce, Poland https://orcid.org/0009-0007-5673-6403, marysiah05@gmail.com Julia Lubomirska, student Jan Kochanowski University, Collegium Medicum, al. IX Wieków Kielc 19A, 25-317 Kielce, Poland https://orcid.org/0009-0008-8557-5108, lubek1999@poczta.onet.pl Patrycja Figurowska, MD Independent Public Healthcare Center in Mińsk Mazowiecki, Szpitalna 37, 05-300 Mińsk Mazowiecki, Poland https://orcid.org/0009-0003-7269-6916, patrycja.figurowska@gmail.com

Łukasz Ciulkiewicz, MD
Independent Public Healthcare Center in Mińsk Mazowiecki, Szpitalna 37, 05-300 Mińsk Mazowiecki, Poland
https://orcid.org/0009-0005-4531-7532, lukasz.ciulkiewicz@onet.eu
Patryk Pluta, MD
Stanisław Rybicki Regional Polyclinical Hospital in Skierniewice, Rybickiego 1, 96-100 Skierniewice, Poland
https://orcid.org/0009-0005-3251-2267, patrykpluta15@gmail.com

*These two authors are equal contributors to this work and designated as co-first authors.

Corresponding author: Arkadiusz Bydliński, student, bydlinskiarkadiusz@gmail.com

ABSTRACT

Introduction

Multiple sclerosis (SM) is the most common chronic inflammatory-demyelinating disease of the Central Nervous System. In its course, a multifocal damage is done to a nervous tissue - myelin sheaths surrounding axons become disintegrated. This process leads to a dysfunction of the electrical impulses passage along nerve fibers which occurs as various symptoms that hinder the functioning of the patients.

Aim of the study

This review aims to summarize treatment methods of relapsing-remitting type of multiple sclerosis (RRMS). Following information will be presented: groups of medications, mechanism of its action, its most important side effects and its effectiveness. Material and method

This review presents the current state of knowledge about relapsing-remitting type of multiple sclerosis (RRMS) treatment - it is based on Pubmed database and was carried out using keywords concerning the issues of multiple sclerosis. Cited articles were selected in order to base on up-to-date investigation results. Summary

A wide range of therapeutic options let neurologists treat their patients accurately, but the basic knowledge about each substance should be assimilated to choose the best individual therapy. Patients suffering from multiple sclerosis differ from each other and a personal approach is needed to implement appropriate treatment as soon as possible. Early start allows patients to not to lose their everyday functioning areas, as the neurodegenerative process is stopped at the initial stage. If some changes in brain tissue structure are permanently done, neurologists should know the options of symptomatic treatment to alleviate a patient's suffering.

Keywords

multiple sclerosis, SM, RRMS, multiple sclerosis treatment, SM treatment, RRMS treatment

INTRODUCTION

Multiple sclerosis (SM) is a neurological syndrome that affected 2.8 million patients in 2020 with its global prevalence of 35.9 per 100,000 people according to the Atlas of MS [1]. It can occur at any age, however the most exposed are patients between 20 and 40 years of

age [2]. Risk of being affected stands at 0.1% in the general population and having an affected first-degree relative increases the risk to 2-4% [3].

Although many risk factors were discovered throughout the years, there was not identified one particular causative agent. Presumably, multiple sclerosis occurs as a result of the series of interactions between genetic predispositions and environmental factors [4]. It has been proven that immunological dysfunctions are involved in the pathophysiology of this syndrome [5], including such immune cells as regulatory T cells (Tregs) [6], T helper cells [7] and B cells [8].

The diagnostic process is based on McDonald criteria (recently modified in 2017) and includes searching for lesions that fulfill requirements of dissemination in time (DIT) and dissemination in space (DIS) - proven by clinical presentation, magnetic resonance imaging (MRI) and the presence of oligoclonal bands in cerebrospinal fluid (CSF) [9].

Observing the course of multiple sclerosis, it can be assigned to one of four typical variants described by The International Advisory Committee on Clinical Trials of MS (also known as Lublin Classification) [10]:

- 1) Clinically isolated syndrome (CIS)
- 2) Relapsing-remitting multiple sclerosis (RRMS)
- 3) Primary progressive multiple sclerosis (PPMS)
- 4) Secondary progressive multiple sclerosis (SPMS)

85-90% of patients suffering from MS present the relapsing-remitting type (RRMS) [11].

AIMS OF TREATMENT

Actions undertaken in order to provide help to patients that are diagnosed with RRMS include:

- 1) treatment of acute relapses
- 2) disease-modifying treatment
- 3) reduction of associated symptoms

Symptomatic attacks worsen the general condition of patients and can be the cause of new signs of disease activity - implementing acute treatment allows patients to suppress immunological reaction and in consequence re-enter (mainly partially) to the state before the attack [12].

Chronic management includes disease-modifying therapy that has significantly developed in recent years. It ought to be introduced as early as possible, as high-efficacy

disease modifying therapies (HE-DMTs) can delay the progression of pathological process the formation of new lesions in the central nervous system. Deceleration in demyelination causes patients' disease develop slower and the number of attacks are limited - in consequence they are able to remain their functioning status for a longer time [13]. Moreover, it prevents intense brain tissue atrophy, which is the cause of progressing patient's destabilization [13]. Therapeutic goals that are measurable include clinical state assessment (number of relapses and their intensity - annualized relapse rate - ARR) and observing disease activity in MRI defined by new/enlarging T2-hyperintense lesions and/or Gd-enhancing lesions [14].

When treating multiple sclerosis as a whole, eliminating the effects of formed lesions should not be ignored. Neurologists' attention has to embrace symptoms like: spasticity, muscle weakness, tremor, pain, bladder dysfunction, constipation, fatigue, paraesthesia, depression, cognitive impairment and sexual dysfunction. If primary treatment of multiple sclerosis is not able to eliminate earlier formed lesions, patients should be additionally treated to decrease dysfunctions hindering their daily functioning. In certain cases, symptomatic treatment is the only way to reduce patients' suffering. Choice of pharmacological and non-pharmacological symptomatic treatment depends on the personal needs of each patient individually [15].

TREATMENT OF ACUTE RELAPSES

Relapses of multiple sclerosis are defined as acute episodes of new neurologic dysfunctions or recurrence of the ones that resolved in the past. They have to be differentiated from pseudo-relapses that can be caused by fever, infection and chronic stress. The management should be focused on three pillars: 1) improvement of functioning of neurons affected by inflammatory demyelination - by expediting their recovery 2) reduction of current attack severity 3) decreasing or fully eliminating clinical symptoms caused by the attack [16].

Corticosteroids (CS) appeared to be an effective choice of therapy of acute MS exacerbations because of their immunosuppressive and immunomodulatory activity [17]. CS were proven to affect matrix metalloproteinases (MMPs) - proteolytic enzymes that take part in cell migration through an extracellular matrix. The fact that CS decrease the level of MMP-9 and increase the level of MMP inhibitors results in strengthening the blood-brain barrier and in consequence decreases the amount of immune cells entering brain tissue [18]. Another effect is related to CS surface receptores occurring on many cells, including immune cells.

The signal triggered by connection between CS and their receptors leads to lymphocytes death by apoptosis [19].

Short-term treatment using high-dose corticosteroids is a considered practice. The most common is parenterally administered methylprednisolone - 1g/day for 3-5 days [17]. Indicated method has been proved to decrease the intensity of exacerbation, improve patients' mobility and promote recovery phase [17]. As the therapy does not last a long time, adverse effects of CS are not frequent. If so, patients may mainly suffer from depression, anxiety, weakness, sleeping disorder, hypertension, gastritis and reflux. Usually neurologists add potassium supplementation and proton pump inhibitors during acute attack treatment to minimize the associated risk.

DISEASE-MODIFYING TREATMENT

Nowadays, the amount of disease-modifying therapies (DMTs) increases. Neurologists can offer patients suffering from MS a wide range of medications that are chosen on the basis of individual factors among which the most important are: age, comorbidities, plans for pregnancy, number and location of lesions, side effects and efficiency [20]. Therapeutics should be used one at a time, the combination therapy was not proven to be more effective [20]. If it is necessary to convert one therapy into another (as its individual efficiency is not satisfactory or adverse effects are intensified), neurologists should always be vigilant on the process of change. The therapy is obviously carried out as a step-up approach, but opposite situations occur too - in some cases a step-down approach is also implemented, especially when a patient's disease is stable for a long time and adverse effects of therapy begin to emerge.

Due to its mechanism of action, DMTs can be divided into 3 categories: [20]

- 1) Immunomodulatory agents
- 2) Continuous lymphocyte depleting agents
- 3) Time-limited immunosuppression followed by immune repopulation

Treatment aiming at various levels of RRMS is a result of numerous studies carried out over the past recent years and allows to choose individual therapy for each patient having regard to their needs.

Effectiveness of each therapy is presented as a reduction in annualized relapses rate (ARR) and reduction of disease activity in MRI (GdE lesions).

1) Immunomodulatory agents

Interferon β -1a, peginterferon β -1a, interferon β -1b - their action mechanisms are based on down-regulation of expression of MHC molecules on antigen-presenting cells, preventing immune cells passage through blood-brain barrier and also influencing on cytokines expressed as decreasing pro-inflammatory and increasing anti-inflammatory ones [21]. Clinical trial results in comparison to placebo: IFN β -1a administered subcutaneously revealed reduction in ARR: 32% and in GdE lesions: 67%; intramuscular administration resulted in reduction in ARR: 18% and in GdE lesions: 32% [22]; PEGylated IFN β-1a resulted in reduction in ARR: 27% and in GdE lesions: 86% [23]; IFN β-1b resulted in reduction in ARR: 34% and in GdE lesions: 83% [24]. Dose and route of administration: IFN β -1a (Rebif) 44 µg 3x/week subcutaneously, IFN β -1a (Avonex) 30 µg 1x/week intramuscularly, PEGylated IFN β -1a 125 µg 1x/2weeks subcutaneously, IFN β -1b 250 µg every other day subcutaneously [25]. Notable adverse effects include mainly flu-like symptoms and injection-site reactions. During IFN therapy neurologists should take care of monitoring such parameters as full blood examination (FBE) and liver function tests (LFTs) at one, three and six months after first dose and then once a year [20]. Some patients may not respond to IFN treatment due to the presence of neutralizing antibodies (NAB) [20].

Glatiramer acetate is a synthetic polypeptide composed of 4 amino acids. Its mechanism of action might involve blocking myelin antigen presentation to T cells [20]. Clinical trial results in comparison to placebo - reduction in ARR: 29% and reduction in GdE lesions was not adequately assessed [26]. Glatiramer acetate is administered subcutaneously, 20 mg 1x/day or 40 mg 3x/week [25]. Adverse effects are not common and are connected with injection. It should be noticed that every patient develops antibodies to glatiramer but it is not associated with less therapeutic effectiveness [20].

Dimethyl fumarate (DMF) is a fumaric acid ester that activates erythroid-derived 2 nuclear factor (NRF2) and consequently has antioxidative activity. It also reduces neuronal excitotoxicity (a process that leads to neuron cells death) and modulates immune cells passage through the blood-brain barrier [27]. Clinical trial results in comparison to placebo - reduction in ARR: 44-53% and reduction in GdE lesions: 74-90% [28]. DMF is administered orally - the titration dose is 120 mg 2x/day x 7 days and the maintenance dose is 240 mg 2x/day [25]. The most common adverse effects are flushing and gastrointestinal symptoms - both of them tend to improve within one month of therapy. Standard monitoring during DMF

treatment should involve full blood examination (FBE) every 3-6 months, liver function tests (LFTs) every 6-12 months and urinalysis once a year [20].

Teriflunomide - an active metabolite of leflunomide - selectively and reversibly inhibits dihydroorotate dehydrogenase and therefore de novo pyrimidine synthesis is suppressed, which results in inhibition of proliferation of activated and autoreactive T and B lymphocytes [25]. Clinical trial results in comparison to placebo - reduction in ARR: 32% and reduction in GdE lesions: 80% [30]. Teriflunomide is administered orally, 14 mg 1x/day [25]. The most common adverse effects include nausea, diarrhea, hair thinning, hepatotoxicity and teratogenicity (applies to both men and women, an effective contraception is necessary) [25]. If needed, a washout therapy with cholestyramine is available - mainly used in case of pregnancy [29]. Standard monitoring during therapy involves full blood examination (FBE) and liver function tests (LFTs) once a month for six months and then every 4-8 weeks and regular measurement of blood pressure [20].

Sphingosine-1-phosphate receptor modulators - a group that includes fingolimod, siponimod, ozanimod and ponesimod. Their mechanism of action is related to a blockade of lymphocyte egression from a lymph node, which causes that the number of central memory and naive T and B lymphocytes is reduced so is their recirculation to the central nervous system [31]. Fingolimod has also an anti-inflammatory and neuroprotective influence due to its interaction with astrocytes and binding with S1P5 receptors on oligodendrocytes [32]. Siponimod and ozanimod are second-generation S1P receptor modulators that bind to S1P1 and S1P5 receptors [33,34]. Preclinical data also suggests that siponimod decreases axonal demyelination and increases axonal remyelination via oligodendrocytes [35]. Ponesimod is the newest one in the S1P receptor modulators group, it highly selectively binds to S1P1 receptor [36]. Clinical trial results in comparison to placebo - fingolimod revealed reduction in ARR: 54% and in GdE lesions: 82% [37]; siponimod revealed reduction in ARR: 55% and in GdE lesions: 82% [38]. Ozanimod revealed reduction in ARR: 48% and in GdE lesions: 63% in comparison to IFN β -1a [39]. Ponesimod revealed reduction in ARR: 31% in comparison to teriflunomide [40]. Every S1P receptor modulators are administered orally: fingolimod 0,5 mg 1x/d, siponimod 2 mg 1x/d after initial uptitration, ozanimod 0,92 mg 1x/d after initial uptitration and ponesimod 20 mg 1x/d after initial uptitration [25]. Adverse effects optionally caused by this therapy include lymphopenia, macular oedema, infections and firstdose bradycardia [25]. Standard monitoring should involve: lymphocyte count, liver function

tests (LFTs), blood pressure measurement and also every patient should have their visual acuity examined and skin cancer screening done every year [20].

Natalizumab is a humanized monoclonal antibody that inhibits $\alpha 4\beta 1$ integrin (an adhesion molecule that is expressed on the lymphocytes' surface) and thereby prevents autoreactive lymphocytes from crossing the blood-brain barrier [41]. Clinical trial results in comparison to placebo - reduction in ARR: 68% and in GdE lesions: 92% [42]. Natalizumab was the first therapy that was approved as an intravenous infusion form - the dose is 300 mg, the procedure of administration lasts over one hour and the patient gets its treatment every 4 weeks [20]. Adverse effects include headache, fatigue or gastrointestinal symptoms. An issue that limits the use of natalizumab is related to progressive multifocal leukoencephalopathy (PML) among patients who were exposed to JCV in the past. In general, the rate of John Cunningham virus (JCV) infection is not rare - more than half of the population is seropositive [43]. Every patient during natalizumab therapy should be checked for PML risk factors, which involves: the presence of JCV antibodies, more than two years of natalizumab therapy and prior use of any immunosuppressive therapy [44]. The risk of PML in patients who fulfill all the three criteria was estimated at 11.1 cases per 1000 patients [44]. A pivotal factor is positive JCV antibody serostatus - it was not proven that greater treatment duration would contribute to a greater PML risk among patients who are seronegative. The antibody titer has a great influence too - according to data, the lowest risk of PML is among patients whose JCV antibody index is ≤ 0.9 and who are on natalizumab treatment for 1-24 months (0.1 PML cases per 1000 patients), when the highest risk is among patients whose index is >1,5 and who are on treatment >48 months (5.4 PML cases per 1000 patients) [45]. In view of the above, standard monitoring involves brain MRI every 6-12 months and JCV antibody testing every 6 months [29].

2) Continuous lymphocyte-depleting agents

Ocrelizumab is a humanized anti-CD20 monoclonal antibody. Antigen CD20 is expressed on the surface of B lymphocytes - its role is to enable proper B-cell immune response. The effects of ocrelizumab involve: selective depletion of CD20-expressing B lymphocytes what halts the preexisting humoral immunity and prevents their reconstitution, blockage of B cells passage from the periphery to the CNS, reduction of antigen presentation by B cells to T cells, reduction of B cells activation and differentiation to forms of plasmablast that secrete immunoglobulins and B cells secretion modulation what reduces the level of proinflammatory cytokines [46, 47]. Clinical trial results in comparison to IFN β-1a reduction in ARR: 46% and 47% and reduction in GdE lesions: 94% and 95%, additional analysis (that was not an endpoint described in study protocol) revealed also a reduction in loss of brain tissue volume [48]. The maintenance dose of ocrelizumab is 300 mg administered intravenously every 6 months after initial uptitration [25]. Combined safety analysis revealed that ocrelizumab is usually well-tolerated, the most appearing adverse effects include infusion-related reactions and neutropenia that can result in mild infections like urinary tract or upper-respiratory tract infections (though some serious infections like pneumonia or cellulitis were also noticed) [49]. Some serious complications were also noticed - there have been several cases of PML in patients on ocrelizumab treatment but most of them were previously on natalizumab treatment [50]. Hepatitis B reactivation can also occur. There was some suspicion of an increased risk of malignancy (<1%) among patients on natalizumab treatment (mainly breast cancer) but further trial-extension suggests that the risk is comparable with a population risk and the routine screening should be age-appropriate [51]. Standard monitoring should involve full blood examination (FBE) and comprehensive metabolic panel (CMP) periodically, HBV panel, immunoglobulin level initially and then periodically [25].

Ofatumumab is the first fully human monoclonal antibody directed against the CD20 molecule on B cells' surface. Its mechanism of action is comparable to ocrelizumab - ofatumumab depletes CD20-expressing B lymphocytes [52]. Clinical trials were conducted in comparison to teriflunomide and the results of the study were as follows - reduction in ARR: 51% and 58% and reduction in GdE lesions: 97% and 94% [53]. Ofatumumab is administered subcutaneously - initial dosing is 20 mg at weeks 0, 1 and 2 and the maintenance dosing is 20 mg 1x/month [25]. Its safety profile is similar to ocrelizumab - treatment can be associated with injected-related reactions and mild infections but there is also a risk of some serious infections or HBV reactivation [25]. As this therapy is quite new, its safety requires long-term monitoring in further studies. Standard monitoring should be focused on the same components as in the case of ocrelizumab [25].

3) Time-limited immunosuppression followed by immune repopulation

Alemtuzumab is a humanized anti-CD52 monoclonal antibody. The CD52 molecule is present on the surface of many B and T lymphocytes. Alemtuzumab causes depletion of CD52 positive cells in the mechanism of antibody-dependent cell-mediated cytolysis (ADCC)

and complement-dependent cytolysis (CDC) [54]. After some time the repopulation process slowly starts - hematopoietic precursor cells are the base for new cells which begin to grow by following a distinct temporal pattern [54]. Clinical trial results in comparison to IFN β -1a - reduction in ARR: 55% and 49% and in GdE lesions: 63% and 61% [55]. Alemtuzumab therapy consists of two courses of intravenous infusions 12 months apart. During the first course a patient is given 12 mg daily for 5 days and in the second course the dose is also 12 mg but the procedure lasts 3 days [25]. If needed, a third course can be provided [25]. Due to the fact that adverse effects of alemtuzumab therapy can be major, patients undergo a strictly monitoring through the Risk Evaluation Mitigation Strategy program, which involves blood tests and urinalysis - its aim is to capture potential autoimmune disorders, as there occurred cases of thyroid dysfunction (30-40% of patients), immune thrombocytopenic purpura or antiglomerular basement membrane disease [20]. The monitoring should also involve annual skin and gynecological cancers screening [25]. It is recommended to undergo Herpes virus prophylaxis - acyclovir 200-400 mg twice daily - from the start of every alemtuzumab course for 2 months (the level of CD4 lymphocytes should be >200 cells/µl) [56].

Cladribine is an analog of deoxyadenosine that mechanism of action is targeted to interfere with DNA synthesis and thereby to deplete immune cells (it preferentially leads to apoptosis of peripheral B and T lymphocytes) [57]. Clinical trial results in comparison to placebo - reduction in ARR: 58% and in GdE lesions: 86% [58]. Cladribine is administered orally in two courses - the cumulative dose is 3.5 mg/kg, so at one cycle the patient gets 1.75 mg/kg [25]. Adverse effects include lymphopenia, so there is a potential risk of infections or reactivation of VZV. During cladribine therapy, patients should have their full blood examination (FBE) with differential checked - before every course, 2 and 6 months after [25]. If the lymphocytes count is <200 cells/mm3, Herpes virus prophylaxis should be considered [20].

REDUCTION OF ASSOCIATED SYNDROMES

Multiple sclerosis is a disease with a wide range of symptoms - nowadays we know that it is mainly determined by the localisation of demyelinating lesions. As the neurodegenerative process may occur in different parts of CNS, symptoms may vary from mild fatigue to complete quadriplegic [10]. Actions undertaken by neurologists in order to alleviate symptoms and facilitate patients' everyday activities should not only be focused on DMTs but also on symptomatic treatment, as not every lesion and its consequences can be cured completely.

Spasticity is presented as an increased muscle stiffness. It is one of the most vexing ailments - according to data, it may affect daily living activities of 44% patients with MS [59]. A very important part of preventing spasticity is avoiding the triggering and aggravating factors which include: infections, stress, constipation, excessive fatigue, fever and certain medicines (some antidepressants and DMTs like interferon) [60]. Non-pharmacological actions that are effective should comprise physiotherapy - it was proven that it decreases the spasticity and also improves the overall response to disease-modifying treatment [61]. Physical exercises are also useful when patients suffer from muscle weakness that is caused by their atrophy. Pharmacological therapies - baclofen, tizanidine, dantrolene and diazepam - are also recommended to be considered as studies have shown their effectiveness [59].

Tremor occurs in 25-60% of MS patients [62]. It can affect only one part of the body or have a severe form that can be highly disabling. So far, none of the studied therapies have been proven to be effective to a satisfactory degree. Some relief can be provided by high doses of isoniazid, propranolol, carbamazepine and glutethimide [62]. Some surgical procedures were reported to decrease tremor - studies were focused on stereotactic thalamotomy and thalamic stimulation - though the groups were small and the long-term functional outcome is not fully known [62].

Many patients with multiple sclerosis have to struggle with chronic pain that dramatically reduces their quality of life. The most common form is chronic neuropathic pain [63]. It leads patients to the next problem, as the treatment offered by modern medicine is not always effective and even if it is, it can face patients with potential adverse effects of another substances they have to take to exist. Using the proposed division, neurologists can offer MS patients three lines of pain-relieving therapy [63]. The first-line treatment includes gabapentin, tricyclic antidepressants, serotonin/norepinephrine reuptake inhibitors (SSRI/SNRI) and topical lidocaine. The second-line therapy involves opioid analgesics or tramadol (alone or combined with adjuvant substances from the first-line therapy). Third-line treatment is based on antiepileptic drugs (lamotrigine, carbamazepine, oxcarbazepine, valproic acid or topiramate), mexiletine and topical capsaicin [63]. The choice of appropriate drug should be in the hands of both doctor and patient to decide which of the following will be the best solution for current problems individually.

Fatigue is a common problem that patients suffering from multiple sclerosis have to face, but it is also a type of issue that entails other struggles. Muscle weakness and decreased endurance make patients feel fatigue faster, so they reduce their physical activity, which exacerbates the problem [64]. Studies showed that aerobic exercise (such as swimming, jogging, cycling), although in some cases very tough for patients, has a great impact on their fatigue level and general functioning. Regularly undertaken physical effort (adjusted for patients' ability) increases muscle strength, endurance and cardio-respiratory fitness, decreases the level of fatigue, improves patients' motor coordination and enables them to maintain their ability in performing daily tasks for a longer time [64].

Depression is a serious problem among patients suffering from neurological diseases, so do multiple sclerosis. Its intensity can be partially reduced by actions undertaken for other purposes (such as physical exercise, as mentioned before, that decreases fatigue) but additional treatment should be implemented if the mental condition of the patient requires it. As a first-line in pharmacological therapy, use of selective serotonin uptake inhibitors (SSRI) is recommended [65]. An extra caution should be exercised when using sedating or anticholinergic drugs, such as tricyclic antidepressants, as they can decrease patients' balance and increase cognitive impairment or bladder problems [65]. Patients' mental well-being can also be supported by attending psychotherapy. Providing patients with cognitive-behavioral therapy (CBT) is proved to be beneficial for them - it helps in correcting distorted perception of environment that the disease brought them, changing their core beliefs so that they are more motivated to participate in everyday life and improving their coping skills, which also impacts on cognitive dysfunctions and chronic fatigue [65].

Cognitive impairment is the issue that hinders daily functioning of many patients suffering from multiple sclerosis. Depending on the studies, the percentage of patients struggling with this problem is 34-65% [66]. The most affected cognitive domains are memory and information processing [67]. The first noticeable disorders depend on the affected area - some may be mild and patients may be able to compensate for it, while some may be very severe and may disturb patients' everyday functioning. Treatment of cognitive dysfunction is not easy, as it is a wide range of deficits. The most important step in the treatment process is an accurate assessment, patients should be examined by a neuropsychologist - specialist in this field. A variety of specially prepared tests can be carried out to capture the cognitive area that is affected by neurodegenerative process. As precise

information is collected, the patient may start a treatment process focused on their individual needs, utilizing therapeutic and behavioral techniques [66].

Sexual dysfunctions (SD) affect many patients with multiple sclerosis. Both men and women may have problems in their sexual sphere. The most common complaints about male SD relate to: erectile dysfunction (50-75%), ejaculatory/orgasmic dysfunction (50%), reduced libido (39%) and anorgasmia (37%), while female SD may involve: difficulty in achieving orgasm (37,1%), vaginal dryness (35,7%) and reduced libido (31,4%) [68]. Sexual dysfunctions should not be ignored, as they are an important part of quality of life (QoL) reported by patients. Pharmacological agents available in SD treatment involve: phosphodiesterase type 5 inhibitors (sildenafil, vardenafil, tadalafil), prostaglandin E1, dopamine agonist (apomorphine), adrenoceptor antagonists (phentolamine, yohimbine) or hormone therapy. Non-pharmacologically, patients can attend psychotherapy or use special items that facilitates their sexual relations [68].

Bladder dysfunction is an issue that is especially embarrassing for MS patients. An interdisciplinary assessment should be conducted by a neurologist and urologist. The problem can be divided into two categories: urinary incontinence and urinary hesitancy. Pharmacological treatment that may be effective include oxybutynin or solifenacin (in case of incontinence) and doxazosin or alfuzosin (in case of hesitancy) [69].

Neurogenic bowel dysfunction is one of the most common problems experienced by MS patients. It comprises two different conditions: constipation (caused by sensorimotor neuropathy that leads to fecal retention as the desire to defecate is decreased) and fecal incontinence (caused by anal sphincter weakness) [70]. Strategies used in order to help patients with constipation are both non-pharmacological and pharmacological. To prevent constipation, patients should firstly increase the amount of liquids and follow a high-fiber diet. Pharmacological agents that were proved to impact on bowel movements involve prucalopride, linaclotide, tegaserod and bisacodyl. Lactulose and bulk laxatives are also effective [70]. Fecal incontinence can also be treated pharmacologically - loperamide and codeine appeared to decrease the symptom of urgency and improved consistency of a stool [71].

CONCLUSIONS

Multiple sclerosis, a disease that years ago was perceived as a death sentence, still remains incurable, but modern medicine offers a wide range of medicines allowing patients to

keep living as usual. Disease-modifying therapies can be chosen individually for every patient according to their needs. Early implementation of treatment stops the progression of MS and thereby makes it possible to stay efficient physically and mentally for many years in most patients. Newer and newer therapeutics appearing on the neurological stage give hope for patients who do not respond to the previous treatment. Deepening the knowledge and understanding the process standing behind multiple sclerosis create new opportunities for next treatment methods.

Disclosures

Author's contribution:

Conceptualization: Arkadiusz Bydliński, Aleksandra Brożyna; Methodology: Natalia Małek, Julia Lubomirska; Formal analysis: Patryk Pluta, Sara Emerla, Anita Kwiatkowska; Investigation: Maria Hermanowska, Patrycja Figurowska; Writing-rough preparation: Łukasz Ciulkiewicz, Konrad Karłowicz; Writing-review and editing: Arkadiusz Bydliński, Aleksandra Brożyna; Supervision: Arkadiusz Bydliński, Aleksandra Brożyna; All authors have read and agreed with the published version of the manuscript. **Funding Statement:** This Research received no external funding. **Institutional Review Board Statement:** Not applicable. **Informed Consent Statement:** Not applicable. **Data Availability Statement:** Not applicable. **Conflicts of Interests:** The authors declare no conflict of interest.

REFERENCES

- Walton C, King R, Rechtman L, Kaye W, Leray E, Marrie RA, Robertson N, La Rocca N, Uitdehaag B, van der Mei I, Wallin M, Helme A, Angood Napier C, Rijke N, Baneke P. Rising prevalence of multiple sclerosis worldwide: Insights from the Atlas of MS, third edition. Mult Scler. 2020 Dec;26(14):1816-1821. doi: 10.1177/1352458520970841. Epub 2020 Nov 11. PMID: 33174475; PMCID: PMC7720355.
- Romero-Pinel L, Bau L, Matas E, León I, Muñoz-Vendrell A, Arroyo P, Masuet-Aumatell C, Martínez-Yélamos A, Martínez-Yélamos S. The age at onset of relapsingremitting multiple sclerosis has increased over the last five decades. Mult Scler Relat Disord. 2022 Dec;68:104103. doi: 10.1016/j.msard.2022.104103. Epub 2022 Aug 9. PMID: 36029708.
- Reich DS, Lucchinetti CF, Calabresi PA. Multiple Sclerosis. N Engl J Med. 2018 Jan 11;378(2):169-180. doi: 10.1056/NEJMra1401483. PMID: 29320652; PMCID: PMC6942519.
- Dendrou CA, Fugger L, Friese MA. Immunopathology of multiple sclerosis. Nat Rev Immunol. 2015 Sep 15;15(9):545-58. doi: 10.1038/nri3871. Epub 2015 Aug 7. PMID: 26250739.
- 5. Dargahi N, Katsara M, Tselios T, Androutsou ME, de Courten M, Matsoukas J, Apostolopoulos V. Multiple Sclerosis: Immunopathology and Treatment Update.

Brain Sci. 2017 Jul 7;7(7):78. doi: 10.3390/brainsci7070078. PMID: 28686222; PMCID: PMC5532591.

- Zozulya AL, Wiendl H. The role of regulatory T cells in multiple sclerosis. Nat Clin Pract Neurol. 2008 Jul;4(7):384-98. doi: 10.1038/ncpneuro0832. Epub 2008 Jun 24. PMID: 18578001.
- Rolla S, Bardina V, De Mercanti S, Quaglino P, De Palma R, Gned D, Brusa D, Durelli L, Novelli F, Clerico M. Th22 cells are expanded in multiple sclerosis and are resistant to IFN-β. J Leukoc Biol. 2014 Dec;96(6):1155-64. doi: 10.1189/jlb.5A0813-463RR. Epub 2014 Aug 5. PMID: 25097195.
- Disanto G, Morahan JM, Barnett MH, Giovannoni G, Ramagopalan SV. The evidence for a role of B cells in multiple sclerosis. Neurology. 2012 Mar 13;78(11):823-32. doi: 10.1212/WNL.0b013e318249f6f0. PMID: 22411958; PMCID: PMC3304944.
- Thompson AJ, Banwell BL, Barkhof F, Carroll WM, Coetzee T, Comi G, Correale J, Fazekas F, Filippi M, Freedman MS, Fujihara K, Galetta SL, Hartung HP, Kappos L, Lublin FD, Marrie RA, Miller AE, Miller DH, Montalban X, Mowry EM, Sorensen PS, Tintoré M, Traboulsee AL, Trojano M, Uitdehaag BMJ, Vukusic S, Waubant E, Weinshenker BG, Reingold SC, Cohen JA. Diagnosis of multiple sclerosis: 2017 revisions of the McDonald criteria. Lancet Neurol. 2018 Feb;17(2):162-173. doi: 10.1016/S1474-4422(17)30470-2. Epub 2017 Dec 21. PMID: 29275977.
- Lublin FD, Reingold SC, Cohen JA, Cutter GR, Sørensen PS, Thompson AJ, Wolinsky JS, Balcer LJ, Banwell B, Barkhof F, Bebo B Jr, Calabresi PA, Clanet M, Comi G, Fox RJ, Freedman MS, Goodman AD, Inglese M, Kappos L, Kieseier BC, Lincoln JA, Lubetzki C, Miller AE, Montalban X, O'Connor PW, Petkau J, Pozzilli C, Rudick RA, Sormani MP, Stüve O, Waubant E, Polman CH. Defining the clinical course of multiple sclerosis: the 2013 revisions. Neurology. 2014 Jul 15;83(3):278-86. doi: 10.1212/WNL.000000000000560. Epub 2014 May 28. PMID: 24871874; PMCID: PMC4117366.
- Freeman L, Longbrake EE, Coyle PK, Hendin B, Vollmer T. High-Efficacy Therapies for Treatment-Naïve Individuals with Relapsing-Remitting Multiple Sclerosis. CNS Drugs. 2022 Dec;36(12):1285-1299. doi: 10.1007/s40263-022-00965-7. Epub 2022 Nov 9. PMID: 36350491; PMCID: PMC9645316.
- 12. Galea I, Ward-Abel N, Heesen C. Relapse in multiple sclerosis. BMJ. 2015 Apr 14;350:h1765. doi: 10.1136/bmj.h1765. PMID: 25872511.
- Filippi M, Amato MP, Centonze D, Gallo P, Gasperini C, Inglese M, Patti F, Pozzilli C, Preziosa P, Trojano M. Early use of high-efficacy disease-modifying therapies makes the difference in people with multiple sclerosis: an expert opinion. J Neurol. 2022 Oct;269(10):5382-5394. doi: 10.1007/s00415-022-11193-w. Epub 2022 May 24. Erratum in: J Neurol. 2022 Dec;269(12):6690-6691. PMID: 35608658; PMCID: PMC9489547.
- 14. Gasperini C, Prosperini L, Tintoré M, Sormani MP, Filippi M, Rio J, Palace J, Rocca MA, Ciccarelli O, Barkhof F, Sastre-Garriga J, Vrenken H, Frederiksen JL, Yousry TA, Enzinger C, Rovira A, Kappos L, Pozzilli C, Montalban X, De Stefano N; the MAGNIMS Study Group. Unraveling treatment response in multiple sclerosis: A clinical and MRI challenge. Neurology. 2019 Jan 22;92(4):180-192. doi: 10.1212/WNL.00000000006810. Epub 2018 Dec 26. PMID: 30587516; PMCID: PMC6345120.
- Esfahani MZ, Ahmadi M, Adibi I. Ontology for Symptomatic Treatment of Multiple Sclerosis. Healthc Inform Res. 2022 Oct;28(4):332-342. doi: 10.4258/hir.2022.28.4.332. Epub 2022 Oct 31. PMID: 36380430; PMCID: PMC9672491.

- 16. Hart FM, Bainbridge J. Current and emerging treatment of multiple sclerosis. Am J Manag Care. 2016 Jun;22(6 Suppl):s159-70. PMID: 27356025.
- Frohman EM, Shah A, Eggenberger E, Metz L, Zivadinov R, Stüve O. Corticosteroids for multiple sclerosis: I. Application for treating exacerbations. Neurotherapeutics. 2007 Oct;4(4):618-26. doi: 10.1016/j.nurt.2007.07.008. PMID: 17920542; PMCID: PMC7479685.
- Rosenberg GA, Dencoff JE, Correa N Jr, Reiners M, Ford CC. Effect of steroids on CSF matrix metalloproteinases in multiple sclerosis: relation to blood-brain barrier injury. Neurology. 1996 Jun;46(6):1626-32. doi: 10.1212/wnl.46.6.1626. PMID: 8649561.
- Liotta LA, Tryggvason K, Garbisa S, Hart I, Foltz CM, Shafie S. Metastatic potential correlates with enzymatic degradation of basement membrane collagen. Nature. 1980 Mar 6;284(5751):67-8. doi: 10.1038/284067a0. PMID: 6243750.
- 20. Travers BS, Tsang BK, Barton JL. Multiple sclerosis: Diagnosis, disease-modifying therapy and prognosis. Aust J Gen Pract. 2022 Apr;51(4):199-206. doi: 10.31128/AJGP-07-21-6103. PMID: 35362004.the of Central Nervous System
- 21. Kieseier BC. The mechanism of action of interferon- β in relapsing multiple sclerosis. CNS Drugs. 2011 Jun 1;25(6):491-502. doi: 10.2165/11591110-00000000-00000. PMID: 21649449.
- 22. Randomised double-blind placebo-controlled study of interferon beta-1a in relapsing/remitting multiple sclerosis. PRISMS (Prevention of Relapses and Disability by Interferon beta-1a Subcutaneously in Multiple Sclerosis) Study Group. Lancet. 1998 Nov 7;352(9139):1498-504. Erratum in: Lancet 1999 Feb 20;353(9153):678. PMID: 9820297.
- 23. Calabresi PA, Kieseier BC, Arnold DL, Balcer LJ, Boyko A, Pelletier J, Liu S, Zhu Y, Seddighzadeh A, Hung S, Deykin A; ADVANCE Study Investigators. Pegylated interferon β-1a for relapsing-remitting multiple sclerosis (ADVANCE): a randomised, phase 3, double-blind study. Lancet Neurol. 2014 Jul;13(7):657-65. doi: 10.1016/S1474-4422(14)70068-7. Epub 2014 Apr 30. PMID: 24794721.
- 24. Interferon beta-1b is effective in relapsing-remitting multiple sclerosis. I. Clinical results of a multicenter, randomized, double-blind, placebo-controlled trial. The IFNB Multiple Sclerosis Study Group. Neurology. 1993 Apr;43(4):655-61. doi: 10.1212/wnl.43.4.655. PMID: 8469318.
- 25. Amin M, Hersh CM. Updates and advances in multiple sclerosis neurotherapeutics. Neurodegener Dis Manag. 2023 Feb;13(1):47-70. doi: 10.2217/nmt-2021-0058. Epub 2022 Oct 31. PMID: 36314777; PMCID: PMC10072078.
- 26. Johnson KP, Brooks BR, Cohen JA, Ford CC, Goldstein J, Lisak RP, Myers LW, Panitch HS, Rose JW, Schiffer RB. Copolymer 1 reduces relapse rate and improves disability in relapsing-remitting multiple sclerosis: results of a phase III multicenter, double-blind placebo-controlled trial. The Copolymer 1 Multiple Sclerosis Study Group. Neurology. 1995 Jul;45(7):1268-76. doi: 10.1212/wnl.45.7.1268. PMID: 7617181.
- Mills EA, Ogrodnik MA, Plave A, Mao-Draayer Y. Emerging Understanding of the Mechanism of Action for Dimethyl Fumarate in the Treatment of Multiple Sclerosis. Front Neurol. 2018 Jan 23;9:5. doi: 10.3389/fneur.2018.00005. PMID: 29410647; PMCID: PMC5787128.
- 28. Fox RJ, Miller DH, Phillips JT, Hutchinson M, Havrdova E, Kita M, Yang M, Raghupathi K, Novas M, Sweetser MT, Viglietta V, Dawson KT; CONFIRM Study Investigators. Placebo-controlled phase 3 study of oral BG-12 or glatiramer in multiple sclerosis. N Engl J Med. 2012 Sep 20;367(12):1087-97. doi:

10.1056/NEJMoa1206328. Erratum in: N Engl J Med. 2012 Oct 25;367(17):1673. PMID: 22992072.

- Hauser SL, Cree BAC. Treatment of Multiple Sclerosis: A Review. Am J Med. 2020 Dec;133(12):1380-1390.e2. doi: 10.1016/j.amjmed.2020.05.049. Epub 2020 Jul 17. PMID: 32682869; PMCID: PMC7704606.
- 30. O'Connor P, Wolinsky JS, Confavreux C, Comi G, Kappos L, Olsson TP, Benzerdjeb H, Truffinet P, Wang L, Miller A, Freedman MS; TEMSO Trial Group. Randomized trial of oral teriflunomide for relapsing multiple sclerosis. N Engl J Med. 2011 Oct 6;365(14):1293-303. doi: 10.1056/NEJMoa1014656. PMID: 21991951.
- 31. Matloubian M, Lo CG, Cinamon G, Lesneski MJ, Xu Y, Brinkmann V, Allende ML, Proia RL, Cyster JG. Lymphocyte egress from thymus and peripheral lymphoid organs is dependent on S1P receptor 1. Nature. 2004 Jan 22;427(6972):355-60. doi: 10.1038/nature02284. PMID: 14737169.
- Groves A, Kihara Y, Chun J. Fingolimod: direct CNS effects of sphingosine 1phosphate (S1P) receptor modulation and implications in multiple sclerosis therapy. J Neurol Sci. 2013 May 15;328(1-2):9-18. doi: 10.1016/j.jns.2013.02.011. Epub 2013 Mar 19. PMID: 23518370; PMCID: PMC3640626.
- Kipp M. Does Siponimod Exert Direct Effects in the Central Nervous System? Cells. 2020 Jul 24;9(8):1771. doi: 10.3390/cells9081771. PMID: 32722245; PMCID: PMC7463861.
- 34. Rasche L, Paul F. Ozanimod for the treatment of relapsing remitting multiple sclerosis. Expert Opin Pharmacother. 2018 Dec;19(18):2073-2086. doi: 10.1080/14656566.2018.1540592. Epub 2018 Nov 8. PMID: 30407868.
- 35. Dietrich M, Hecker C, Martin E, Langui D, Gliem M, Stankoff B, Lubetzki C, Gruchot J, Göttle P, Issberner A, Nasiri M, Ramseier P, Beerli C, Tisserand S, Beckmann N, Shimshek D, Petzsch P, Akbar D, Levkau B, Stark H, Köhrer K, Hartung HP, Küry P, Meuth SG, Bigaud M, Zalc B, Albrecht P. Increased Remyelination and Proregenerative Microglia Under Siponimod Therapy in Mechanistic Models. Neurol Neuroimmunol Neuroinflamm. 2022 Mar 30;9(3):e1161. doi: 10.1212/NXI.00000000001161. PMID: 35354603; PMCID: PMC8969301.
- 36. D'Ambrosio D, Freedman MS, Prinz J. Ponesimod, a selective S1P1 receptor modulator: a potential treatment for multiple sclerosis and other immune-mediated diseases. Ther Adv Chronic Dis. 2016 Jan;7(1):18-33. doi: 10.1177/2040622315617354. PMID: 26770667; PMCID: PMC4707431.
- 37. Kappos L, Radue EW, O'Connor P, Polman C, Hohlfeld R, Calabresi P, Selmaj K, Agoropoulou C, Leyk M, Zhang-Auberson L, Burtin P; FREEDOMS Study Group. A placebo-controlled trial of oral fingolimod in relapsing multiple sclerosis. N Engl J Med. 2010 Feb 4;362(5):387-401. doi: 10.1056/NEJMoa0909494. Epub 2010 Jan 20. PMID: 20089952.
- 38. Kappos L, Bar-Or A, Cree BAC, Fox RJ, Giovannoni G, Gold R, Vermersch P, Arnold DL, Arnould S, Scherz T, Wolf C, Wallström E, Dahlke F; EXPAND Clinical Investigators. Siponimod versus placebo in secondary progressive multiple sclerosis (EXPAND): a double-blind, randomised, phase 3 study. Lancet. 2018 Mar 31;391(10127):1263-1273. doi: 10.1016/S0140-6736(18)30475-6. Epub 2018 Mar 23. Erratum in: Lancet. 2018 Nov 17;392(10160):2170. PMID: 29576505.
- 39. Comi G, Kappos L, Selmaj KW, Bar-Or A, Arnold DL, Steinman L, Hartung HP, Montalban X, Kubala Havrdová E, Cree BAC, Sheffield JK, Minton N, Raghupathi K, Ding N, Cohen JA; SUNBEAM Study Investigators. Safety and efficacy of ozanimod versus interferon beta-1a in relapsing multiple sclerosis (SUNBEAM): a multicentre, randomised, minimum 12-month, phase 3 trial. Lancet Neurol. 2019

Nov;18(11):1009-1020. doi: 10.1016/S1474-4422(19)30239-X. Epub 2019 Sep 3. PMID: 31492651.

- 40. Kappos L, Fox RJ, Burcklen M, Freedman MS, Havrdová EK, Hennessy B, Hohlfeld R, Lublin F, Montalban X, Pozzilli C, Scherz T, D'Ambrosio D, Linscheid P, Vaclavkova A, Pirozek-Lawniczek M, Kracker H, Sprenger T. Ponesimod Compared With Teriflunomide in Patients With Relapsing Multiple Sclerosis in the Active-Comparator Phase 3 OPTIMUM Study: A Randomized Clinical Trial. JAMA Neurol. 2021 May 1;78(5):558-567. doi: 10.1001/jamaneurol.2021.0405. PMID: 33779698; PMCID: PMC8008435.
- 41. Yednock TA, Cannon C, Fritz LC, Sanchez-Madrid F, Steinman L, Karin N. Prevention of experimental autoimmune encephalomyelitis by antibodies against alpha 4 beta 1 integrin. Nature. 1992 Mar 5;356(6364):63-6. doi: 10.1038/356063a0. PMID: 1538783.
- 42. Polman CH, O'Connor PW, Havrdova E, Hutchinson M, Kappos L, Miller DH, Phillips JT, Lublin FD, Giovannoni G, Wajgt A, Toal M, Lynn F, Panzara MA, Sandrock AW; AFFIRM Investigators. A randomized, placebo-controlled trial of natalizumab for relapsing multiple sclerosis. N Engl J Med. 2006 Mar 2;354(9):899-910. doi: 10.1056/NEJMoa044397. PMID: 16510744.
- Pardo G, Jones DE. The sequence of disease-modifying therapies in relapsing multiple sclerosis: safety and immunologic considerations. J Neurol. 2017 Dec;264(12):2351-2374. doi: 10.1007/s00415-017-8594-9. Epub 2017 Sep 6. Erratum in: J Neurol. 2017 Dec;264(12):2375-2377. PMID: 28879412; PMCID: PMC5688209.
- 44. Bloomgren G, Richman S, Hotermans C, Subramanyam M, Goelz S, Natarajan A, Lee S, Plavina T, Scanlon JV, Sandrock A, Bozic C. Risk of natalizumab-associated progressive multifocal leukoencephalopathy. N Engl J Med. 2012 May 17;366(20):1870-80. doi: 10.1056/NEJMoa1107829. PMID: 22591293.
- 45. Plavina T, Subramanyam M, Bloomgren G, Richman S, Pace A, Lee S, Schlain B, Campagnolo D, Belachew S, Ticho B. Anti-JC virus antibody levels in serum or plasma further define risk of natalizumab-associated progressive multifocal leukoencephalopathy. Ann Neurol. 2014 Dec;76(6):802-12. doi: 10.1002/ana.24286. Epub 2014 Oct 24. PMID: 25273271; PMCID: PMC4282070.
- 46. Li R, Patterson KR, Bar-Or A. Reassessing B cell contributions in multiple sclerosis. Nat Immunol. 2018 Jul;19(7):696-707. doi: 10.1038/s41590-018-0135-x. Epub 2018 Jun 20. PMID: 29925992.
- 47. Sabatino JJ Jr, Pröbstel AK, Zamvil SS. B cells in autoimmune and neurodegenerative central nervous system diseases. Nat Rev Neurosci. 2019 Dec;20(12):728-745. doi: 10.1038/s41583-019-0233-2. Epub 2019 Nov 11. Erratum in: Nat Rev Neurosci. 2020 Jan;21(1):56. PMID: 31712781.
- 48. Hauser SL, Bar-Or A, Comi G, Giovannoni G, Hartung HP, Hemmer B, Lublin F, Montalban X, Rammohan KW, Selmaj K, Traboulsee A, Wolinsky JS, Arnold DL, Klingelschmitt G, Masterman D, Fontoura P, Belachew S, Chin P, Mairon N, Garren H, Kappos L; OPERA I and OPERA II Clinical Investigators. Ocrelizumab versus Interferon Beta-1a in Relapsing Multiple Sclerosis. N Engl J Med. 2017 Jan 19;376(3):221-234. doi: 10.1056/NEJMoa1601277. Epub 2016 Dec 21. PMID: 28002679.
- 49. Hauser SL, Kappos L, Arnold DL, Bar-Or A, Brochet B, Naismith RT, Traboulsee A, Wolinsky JS, Belachew S, Koendgen H, Levesque V, Manfrini M, Model F, Hubeaux S, Mehta L, Montalban X. Five years of ocrelizumab in relapsing multiple sclerosis: OPERA studies open-label extension. Neurology. 2020 Sep 29;95(13):e1854-e1867.

doi: 10.1212/WNL.000000000010376. Epub 2020 Jul 20. PMID: 32690791; PMCID: PMC7682822.

- 50. Lünemann JD, Diaz-Diaz J, Stankoff B, Young C, Siva A, Miller A, Lubetzki C, Wiendl H, Oreja-Guevara C. Highlights from the 2019 European Congress on Treatment and Research in Multiple Sclerosis (ECTRIMS 2019). Mult Scler. 2020 Jun;26(7):859-868. doi: 10.1177/1352458520918377. Epub 2020 May 4. PMID: 32364431.
- 51. Hauser SL, Kappos L, Montalban X, Craveiro L, Chognot C, Hughes R, Koendgen H, Pasquarelli N, Pradhan A, Prajapati K, Wolinsky JS. Safety of Ocrelizumab in Patients With Relapsing and Primary Progressive Multiple Sclerosis. Neurology. 2021 Oct 19;97(16):e1546-e1559. doi: 10.1212/WNL.000000000012700. Epub 2021 Sep 2. PMID: 34475123; PMCID: PMC8548959.
- 52. LiverTox: Clinical and Research Information on Drug-Induced Liver Injury [Internet]. Bethesda (MD): National Institute of Diabetes and Digestive and Kidney Diseases; 2012–. Ofatumumab. 2020 Apr 15. PMID: 31643960.
- 53. Hauser SL, Bar-Or A, Cohen JA, Comi G, Correale J, Coyle PK, Cross AH, de Seze J, Leppert D, Montalban X, Selmaj K, Wiendl H, Kerloeguen C, Willi R, Li B, Kakarieka A, Tomic D, Goodyear A, Pingili R, Häring DA, Ramanathan K, Merschhemke M, Kappos L; ASCLEPIOS I and ASCLEPIOS II Trial Groups. Ofatumumab versus Teriflunomide in Multiple Sclerosis. N Engl J Med. 2020 Aug 6;383(6):546-557. doi: 10.1056/NEJMoa1917246. PMID: 32757523.
- 54. Ruck T, Bittner S, Wiendl H, Meuth SG. Alemtuzumab in Multiple Sclerosis: Mechanism of Action and Beyond. Int J Mol Sci. 2015 Jul 20;16(7):16414-39. doi: 10.3390/ijms160716414. PMID: 26204829; PMCID: PMC4519957.
- 55. Cohen JA, Coles AJ, Arnold DL, Confavreux C, Fox EJ, Hartung HP, Havrdova E, Selmaj KW, Weiner HL, Fisher E, Brinar VV, Giovannoni G, Stojanovic M, Ertik BI, Lake SL, Margolin DH, Panzara MA, Compston DA; CARE-MS I investigators. Alemtuzumab versus interferon beta 1a as first-line treatment for patients with relapsing-remitting multiple sclerosis: a randomised controlled phase 3 trial. Lancet. 2012 Nov 24;380(9856):1819-28. doi: 10.1016/S0140-6736(12)61769-3. Epub 2012 Nov 1. PMID: 23122652.
- 56. Epstein DJ, Dunn J, Deresinski S. Infectious Complications of Multiple Sclerosis Therapies: Implications for Screening, Prophylaxis, and Management. Open Forum Infect Dis. 2018 Jul 16;5(8):ofy174. doi: 10.1093/ofid/ofy174. PMID: 30094293; PMCID: PMC6080056.
- Leist TP, Weissert R. Cladribine: mode of action and implications for treatment of multiple sclerosis. Clin Neuropharmacol. 2011 Jan-Feb;34(1):28-35. doi: 10.1097/WNF.0b013e318204cd90. PMID: 21242742.
- 58. Giovannoni G, Comi G, Cook S, Rammohan K, Rieckmann P, Soelberg Sørensen P, Vermersch P, Chang P, Hamlett A, Musch B, Greenberg SJ; CLARITY Study Group. A placebo-controlled trial of oral cladribine for relapsing multiple sclerosis. N Engl J Med. 2010 Feb 4;362(5):416-26. doi: 10.1056/NEJMoa0902533. Epub 2010 Jan 20. PMID: 20089960.
- 59. Rizzo MA, Hadjimichael OC, Preiningerova J, Vollmer TL. Prevalence and treatment of spasticity reported by multiple sclerosis patients. Mult Scler. 2004 Oct;10(5):589-95. doi: 10.1191/1352458504ms10850a. PMID: 15471378.
- Meca-Lallana JE, Hernández-Clares R, Carreón-Guarnizo E. Spasticity in multiple sclerosis and role of glatiramer acetate treatment. Brain Behav. 2015 Sep;5(9):e00367. doi: 10.1002/brb3.367. Epub 2015 Jul 14. PMID: 26445705; PMCID: PMC4589813.

- 61. Giovannelli M, Borriello G, Castri P, Prosperini L, Pozzilli C. Early physiotherapy after injection of botulinum toxin increases the beneficial effects on spasticity in patients with multiple sclerosis. Clin Rehabil. 2007 Apr;21(4):331-7. doi: 10.1177/0269215507072772. PMID: 17613573.
- Koch M, Mostert J, Heersema D, De Keyser J. Tremor in multiple sclerosis. J Neurol. 2007 Feb;254(2):133-45. doi: 10.1007/s00415-006-0296-7. Epub 2007 Feb 21. PMID: 17318714; PMCID: PMC1915650.
- 63. Murphy KL, Bethea JR, Fischer R. Neuropathic Pain in Multiple Sclerosis—Current Therapeutic Intervention and Future Treatment Perspectives. In: Zagon IS, McLaughlin PJ, editors. Multiple Sclerosis: Perspectives in Treatment and Pathogenesis [Internet]. Brisbane (AU): Codon Publications; 2017 Nov 27. Chapter 4. PMID: 29261265.
- 64. Razazian N, Kazeminia M, Moayedi H, Daneshkhah A, Shohaimi S, Mohammadi M, Jalali R, Salari N. The impact of physical exercise on the fatigue symptoms in patients with multiple sclerosis: a systematic review and meta-analysis. BMC Neurol. 2020 Mar 13;20(1):93. doi: 10.1186/s12883-020-01654-y. PMID: 32169035; PMCID: PMC7068865.
- 65. Skokou M, Soubasi E, Gourzis P. Depression in multiple sclerosis: a review of assessment and treatment approaches in adult and pediatric populations. ISRN Neurol. 2012;2012:427102. doi: 10.5402/2012/427102. Epub 2012 Oct 14. PMID: 23097716; PMCID: PMC3477767.
- 66. Kalb R, Beier M, Benedict RH, Charvet L, Costello K, Feinstein A, Gingold J, Goverover Y, Halper J, Harris C, Kostich L, Krupp L, Lathi E, LaRocca N, Thrower B, DeLuca J. Recommendations for cognitive screening and management in multiple sclerosis care. Mult Scler. 2018 Nov;24(13):1665-1680. doi: 10.1177/1352458518803785. Epub 2018 Oct 10. PMID: 30303036; PMCID: PMC6238181.
- 67. Benedict RH, DeLuca J, Phillips G, LaRocca N, Hudson LD, Rudick R; Multiple Sclerosis Outcome Assessments Consortium. Validity of the Symbol Digit Modalities Test as a cognition performance outcome measure for multiple sclerosis. Mult Scler. 2017 Apr;23(5):721-733. doi: 10.1177/1352458517690821. Epub 2017 Feb 16. PMID: 28206827; PMCID: PMC5405816.
- Guo ZN, He SY, Zhang HL, Wu J, Yang Y. Multiple sclerosis and sexual dysfunction. Asian J Androl. 2012 Jul;14(4):530-5. doi: 10.1038/aja.2011.110. Epub 2012 Mar 26. PMID: 22447199; PMCID: PMC3720075.
- 69. Phé V, Chartier-Kastler E, Panicker JN. Management of neurogenic bladder in patients with multiple sclerosis. Nat Rev Urol. 2016 May;13(5):275-88. doi: 10.1038/nrurol.2016.53. Epub 2016 Mar 31. PMID: 27030526.
- Gulick EE. Neurogenic Bowel Dysfunction Over the Course of Multiple Sclerosis: A Review. Int J MS Care. 2022 Sep-Oct;24(5):209-217. doi: 10.7224/1537-2073.2021-007. Epub 2022 Jun 20. PMID: 36090242; PMCID: PMC9461721.
- Palmer KR, Corbett CL, Holdsworth CD. Double-blind cross-over study comparing loperamide, codeine and diphenoxylate in the treatment of chronic diarrhea. Gastroenterology. 1980 Dec;79(6):1272-5. PMID: 7002706.